

Kidney Transplantation in Patients With AA Amyloidosis: Outcomes in a French Multicenter Cohort

Chloë Schwarz, Sophie Georgin-Lavialle, Yannis Lombardi, Olivier Marion, Frédéric Jambon, Christophe Legendre, David Marx, Charlène Levi, Fatouma Toure, Moglie Le Quintrec, et al.

▶ To cite this version:

Chloë Schwarz, Sophie Georgin-Lavialle, Yannis Lombardi, Olivier Marion, Frédéric Jambon, et al.. Kidney Transplantation in Patients With AA Amyloidosis: Outcomes in a French Multicenter Cohort. American Journal of Kidney Diseases, 2023, 83 (3), pp.329-339. 10.1053/j.ajkd.2023.07.020 . hal-04220868

HAL Id: hal-04220868 https://u-picardie.hal.science/hal-04220868v1

Submitted on 8 Nov 2024

HAL is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers. L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution 4.0 International License

Kidney Transplantation in Patients With AA Amyloidosis: Outcomes in a French Multicenter Cohort

Chloë Schwarz, Sophie Georgin-Lavialle, Yannis Lombardi, Olivier Marion, Frédéric Jambon, Christophe Legendre, David Marx, Charlène Levi, Fatouma Toure, Moglie Le Quintrec, Mickael Bobot, Marie Matignon, Amaury Dujardin, Mehdi Maanaoui, Sébastien Cuozzo, Arwa Jalal-Eddine, Kévin Louis, Inna Mohamadou, François Brazier, Tristan De Nattes, Claire Geneste, Eric Thervet, Didier Ducloux, Valentin Mayet, Raphaël Kormann, Antoine Lanot, Agnès Duveau, Mohamad Zaidan, Laurent Mesnard, Nacera Ouali, Eric Rondeau, Camille Petit-Hoang, Vincent Audard, Aurélie Deshayes, Anissa Moktefi, Marion Rabant, David Buob, Hélène François, and Yosu Luque

Rationale & Objective: Outcomes of kidney transplantation for patients with renal AA amyloidosis are uncertain, with reports of poor survival and high rates of disease recurrence. However, the data are inconclusive and mostly based on studies from the early 2000s and earlier.

Study Design: Retrospective multicenter cohort study.

Setting & Participants: We searched the French national transplant database to identify all patients with renal AA amyloidosis who underwent kidney transplantation between 2008 and 2018.

Exposures: Age, cause of amyloidosis, use of biotherapies, and C-reactive protein levels.

Outcomes: Outcomes were all-cause mortality and allograft loss. We also reported amyloidosis allograft recurrence, occurrence of acute rejection episodes, as well as infectious, cardiovascular, and neoplastic disease events.

Analytical Approach: Kaplan-Meier estimator for mortality and cumulative incidence function method for allograft loss. Factors associated with patient and allograft survival were investigated using a Cox proportional hazards model and a cause-specific hazards model, respectively.

Results: 86 patients who received kidney transplants for AA amyloidosis at 26 French centers

were included. The median age was 49.4 years (IQR, 39.7-61.1). The main cause of amyloidosis was familial Mediterranean fever (37 cases; 43%). 16 (18.6%) patients received biotherapy after transplantation. Patient survival rates were 94.0% (95% Cl, 89.1-99.2) at 1 year and 85.5% (77.8-94.0) at 5 years after transplantation. Cumulative incidences of allograft loss were 10.5% (4.0-17.0) at 1 year and 13.0% (5.8-20.1) at 5 years after transplantation. Histologically proven AA amyloidosis recurrence occurred in 5 transplants (5.8%). An infection requiring hospitalization developed in 55.8% of cases, and there was a 27.9% incidence of acute allograft rejection. Multivariable analysis showed that C-reactive protein concentration at the time of transplantation was associated with patient survival (HR, 1.01; 95% CI, 1.00-1.02; P = 0.01) and allograft survival (HR, 1.68; 95% Cl, 1.10-2.57; P = 0.02).

Limitations: The study lacked a control group, and the effect of biotherapies on transplantation outcomes could not be explored.

Conclusions: This relatively contemporary cohort of patients who received a kidney transplant for AA amyloidosis experienced favorable rates of survival and lower recurrence rates than previously reported. These data support the practice of treating these patients with kidney transplantation for end-stage kidney disease. Complete author and article information provided before references.

Correspondence to Y. Luque (yosu.luque@ aphp.fr)

Am J Kidney Dis. 83(3):329-339. Published online September 22, 2023.

doi: 10.1053/ j.ajkd.2023.07.020

© 2023 The Authors. Published by Elsevier Inc. on behalf of the National Kidney Foundation, Inc. This is an open access article under the CC BY license (http://creativecommons. org/licenses/by/4.0/).

A myloidosis is characterized by extracellular deposition of amyloid, an insoluble fibril formed by the accumulation of misfolded proteins in a particular β -sheet conformation. Deposition can occur in all tissues and leads to progressive organ dysfunction. Amyloidosis is a heterogenous group of diseases encompassing more than 30 different types, depending on the precursor protein.¹ After wild-type transthyretin cardiac amyloidosis and immunoglobulin light-chain amyloidosis, AA amyloidosis is the most frequent systemic amyloidosis in epidemiologic studies.²⁻⁴ It results from the accumulation of serum amyloid A (SAA), the production of which is mediated by inflammatory cytokines. SAA may accumulate in all

chronic inflammatory states, which can be secondary to infections, inflammatory rheumatological or intestinal diseases, and monogenic autoinflammatory diseases such as familial Mediterranean fever (FMF). Kidney involvement is frequent in AA amyloidosis, and, in the largest series to date of 374 patients with AA amyloidosis, it led to kidney failure in 44% of cases and was the major prognostic factor.⁵

Survival of patients with AA amyloidosis undergoing long-term dialysis is poor.^{6,7} Kidney transplantation may be considered in these patients, but concerns remain regarding increased mortality and the possibility of amyloidosis recurrence in the graft. In a retrospective



PLAIN-LANGUAGE SUMMARY

AA amyloidosis is a severe and rare disease. Kidney involvement is frequent and leads to end-stage kidney disease. Because of the involvement of other organs, these patients are often frail, which has raised concerns about their suitability for kidney transplantation. We reviewed all patients with AA amyloidosis nephropathy who underwent kidney transplantation in France in the recent era (2008-2018) and found that the outcomes after kidney transplantation were favorable, with 85.5% of patients still alive 5 years after transplantation, a survival rate that is comparable to the outcomes of patients receiving a transplant for other forms of kidney diseases. Recurrence of amyloidosis in the transplanted kidney was infrequent (5.8%). These data support the practice of kidney transplantation for patients with AA amyloidosis who experience kidney failure.

French multicenter study, Kofman et al reported outcomes of 59 patients with AA amyloidosis who underwent kidney transplantation between 1981 and 2009 and found a less favorable overall survival, with frequent infections. Disease recurrence occurred in 14% of cases and was associated with poorer outcomes.⁸ Other studies are consistent with these findings and report poor survival and recurrence rates between 8% and 26%.⁹⁻¹⁵ In a recent series by Law et al, survival of AA amyloidosis kidney transplant recipients (N = 48) was comparable to that in controls with diabetic nephropathy.¹⁵ Overall, series regarding kidney transplantation in patients with AA amyloidosis are rather small and heterogenous. There is also a lack of recent data, and all the cited studies had very long inclusion periods.

To date, treatment of AA amyloidosis relies on treating the underlying inflammatory process to prevent further amyloid deposition. In patients with FMF, colchicine has historic efficacy in preventing amyloidosis,¹⁶ including in the setting of kidney transplantation.¹⁷ In the recent era (since 2010), new anti-inflammatory treatments known as biotherapies have been more widely used in systemic autoinflammatory diseases, namely anti–interleukin (IL) 1 and anti–IL-6 therapies: they proved effective in the control of symptoms and systemic inflammation.¹⁸ Similarly, in chronic inflammatory rheumatic and inflammatory bowel diseases, anti–tumor necrosis factor (TNF)- α therapies drastically changed disease prognosis and may reduce amyloidosis occurrence.^{19,20}

Considering these observations, we hypothesized that kidney transplantation for patients with AA amyloidosis may have better outcomes in the era of such biotherapies. We thus conducted a retrospective multicenter cohort study focusing on the recent period of 2008-2018: all patients with AA amyloidosis who underwent kidney transplantation in France during this time were included. The objectives of the present study are to describe the cohort (cause of AA amyloidosis, treatments received), report transplantation outcomes, and try to identify factors associated with patient and graft survival.

Methods

Patients

To identify all patients with AA amyloidosis who underwent kidney transplantation during the period of interest, we searched the French national transplant database CRISTAL (Agence de la Biomédecine) using the criteria "nephropathy = amyloidosis" and "transplantation period = 07/ 01/2008-06/30/2018". We then contacted each transplant center, and medical records of identified cases were reviewed. Inclusion criteria were age of at least 18 years at the time of transplantation, a diagnosis of AA amyloidosis nephropathy, and a kidney transplantation during the period of interest. Renal AA amyloidosis diagnosis was confirmed histologically by a kidney biopsy or by the presence of AA amyloid deposition in another tissue and the coexistence of nephropathy. Exceptions were made for cases of proven FMF with a history of progressive proteinuric kidney disease without another plausible cause. If a patient received more than one transplant during the period of interest, each transplant was included as a separate case.

Data

Patients' medical files were examined retrospectively, and data were recorded anonymously regarding initial characteristics, characteristics of transplantation, and outcomes after transplantation.

Initial characteristics such as demographic data, C-reactive protein (CRP) level at the time of transplantation, cardiovascular disease and diabetes, cause of AA amyloidosis, etiologic treatments before transplantation, duration of dialysis, and number of prior transplantations were recorded.

Characteristics of transplantation included combined transplant, type and age of the donor, cold ischemia time, immunologic characteristics (presence of anti-human leukocyte antigen antibodies and anti-human leukocyte antigen donor-specific antibodies), and immunosuppressive regimen.

Outcomes after transplantation included all-cause mortality and graft failure at 1 and 5 years; occurrence of a biopsy-proven acute graft rejection, an infection requiring hospitalization, an opportunistic infection (according to the definition by Fishman et al in 2007^{21,22}), cancer, or a cardiovascular event (myocardial infarction, acute congestive heart failure, pulmonary embolism, ischemic stroke) during follow-up and the time of occurrence for each event; urinary protein-creatinine ratio (UPCR) and glomerular filtration rate estimated by the MDRD (Modification of Diet in Renal Disease) Study formula at different times (3 months, 1 year, and 5 years after transplantation and at last follow-up); and CRP levels annually during follow-up. Of note, SAA levels were not available for most patients, and CRP was therefore chosen



Figure 1. Flow chart presenting the inclusion process of transplants in the study. Data are presented as number of transplants unless otherwise specified. Abbreviations: AL, immunoglobin light chain; ESKD, end-stage kidney disease; KT, kidney transplant; TTR, transthyretin.

as a surrogate marker because its reliability in the scope of amyloidosis has been established.^{23,24} Administration of a biotherapy was also recorded.

Immunoprophylaxis varied according to each center's practice. In regard to pneumocystosis prevention, the general strategy was to prescribe cotrimoxazole for a minimum of 12 months in the absence of adverse effects; the drug was then continued or stopped depending on the physician's preference. There was no uniform protocol among the different centers in regard to cytomegalovirus prophylaxis.

During follow-up, kidney graft biopsies were performed according to each local center's practice. There was no systematic biopsy during follow-up at some centers, whereas others performed protocol biopsies at 3 months after transplantation, at 12 months after transplantation, or both. All biopsy specimens were analyzed locally, and evidence of Congo red stain analysis was recorded. When possible, biopsies were reanalyzed and additional Congo red stain was performed if missing. Recurrence of AA amyloidosis in the graft was defined by positive Congo red staining with apple-green birefringence under polarized light and anti-SAA immunohistochemical staining of amyloid deposits. The duration of follow-up was recorded, and follow-up was continued until July 1, 2020.

Statistical Analysis

Quantitative data are presented as medians and IQRs. Categorial data are presented as counts and percentages. Survival functions of time-to-event outcomes were estimated using the Kaplan-Meier estimator (for mortality) or the cumulative incidence function method (for graft loss) and compared using the log-rank test or Gray's test, respectively. Missing data were not imputed, and the proportions of missing data are shown in the tables. Factors associated with patient survival and death-censored graft survival were investigated using a Cox proportional hazards model and a cause-specific hazards model, respectively. Studied factors were selected among initial characteristics for their clinical relevance.

A P value <0.05 was considered statistically significant. All statistical tests were performed using R software (version 2020; R Foundation for Statistical Computing).

Ethical Statement

Kidney donation followed the 2008 Declaration of Istanbul principles and the French Agence de la Biomédecine regulations. All patients signed informed written consent for the use of part of the biopsy specimen for scientific purposes. All procedures and the use of tissues were performed in accordance with the principles of the Declaration of Helsinki. According to French regulations (Jardé law), observational studies involving no intervention do not require institutional ethical approval, so none was sought for the present study (General Data Protection Regulation number 2216254; Commission nationale de l'informatique et des libertés).

Results

A total of 168 kidney transplants in 31 centers met the aforementioned search criteria. After contacting each center, 159 transplants were analyzed, and 86 were identified as cases of AA amyloidosis, accounting for 82

AJKD

Table 1. Patient Characteristics

Characteristic	AA Amyloidosis Transplants (n = 86)	Missing Data
Age at KT, y	49.4 (39.7-61.1)	0
Female sex	32 (37.2%)	0
BMI, kg/m ²	22.2 (20.0-24.3)	8 (9.3%)
Preexisting diabetes	3 (3.5%)	0
Preexisting hypertension	45 (52.3%)	4 (4.7%)
Geographic origin		4 (4.7%)
Western Europe	25 (29.1%)	
Turkey, Georgia, Armenia	27 (31.4%)	
Northern Africa	30 (34.9%)	
Cause of amyloidosis		0
FMF	37 (43.0%)	
Other genetic autoinflammatory syndrome	6 (7.0%)	
Chronic inflammatory disease	19 (22.1%)	
Chronic infection	11 (12.8%)	
Unknown	13 (15.1%)	
CRP at KT, mg/L	4.4 (2.8-15.2)	18 (20.9%)
Duration of dialysis before KT, mo	39.5 (19.0-63.8)	0
Second or third KT	9 (10.5%)	0
Age of donor, y	53 (39-65)	5 (5.8%)
Living donor	4 (4.7%)	0
Cold ischemia, min	901 (700-1,200)	4 (4.7%)
Anti-HLA antibodies at KT	38 (44.2%)	10 (11.6%)
DSA at KT	10 (11.6%)	4 (4.7%)
Anti-thymocyte induction	41 (47.7%)	3 (3.5%)
Length of follow-up, mo	55.8 (28.3-86.8)	0

Quantitative data are presented as median (IQR) and categorial data as n (%). Abbreviations: BMI, body mass index; CRP, C-reactive protein; DSA, donorspecific antibody; FMF, familial Mediterranean fever; HLA, human leukocyte antigen; KT, kidney transplantation.

patients in 26 centers. Details of all amyloidosis types are shown in a flow chart in Fig 1.

Patients' Characteristics

Initial characteristics of patients are shown in Table 1. Median age at the time of transplantation was 49.4 (39.7-61.1) years, and median body mass index was 22.2 (20.0-24.3) kg/m². Three patients (3.5%) had preexisting diabetes, and 45 (52.3%) had preexisting hypertension. Histological diagnosis of AA amyloidosis was made by a kidney biopsy in 47 cases (54.7%) and by another tissue biopsy in 31 cases (36.0%); in the remaining 8 cases (9.3%), histological evidence of AA amyloidosis was not found (these were exclusively patients with FMF with no other plausible cause of nephropathy). Causes of amyloidosis were FMF (37 transplants [43.0%]), another autoinflammatory syndrome (6 transplants [7.0%], including TNF receptor–associated periodic syndrome, hyper–immunoglobulin D syndrome, Schnitzler's syndrome, and cryopyrin-associated periodic syndrome), chronic inflammatory disease (19 transplants [22.1%], including Crohn's disease [n = 9], ankylosing spondylitis [n = 4], rheumatoid arthritis [n = 2], Castleman's disease [n = 2], Rosai-Dorfman disease [n = 1], and the association of Crohn's disease with ankylosing spondylitis [n = 1]), and chronic infection (11 transplants [12.8%], including mycobacterium tuberculosis infection [n = 5], bronchiectasis [n = 3], cystic fibrosis [n = 1], Hirschsprung's disease with chronic perineal abscesses [n = 1] and chronic sacral pressure ulcer [n = 1]); the cause of AA amyloidosis was unknown for the remaining 13 cases (15.3%).

Before transplantation, 44 patients (51.2%) received colchicine, 13 (15.1%) anti–IL-1 therapy, 3 (3.5%) anti–IL-6 therapy, and 4 (4.7%) anti–TNF- α therapy. After transplantation, 33 patients (38.4%) received colchicine, 12 (14.0%) anti–IL-1 therapy, 1 (1.2%) anti–IL-6 therapy, and 3 (3.5%) anti–TNF- α therapy. A total of 16 patients (18.6%) received biotherapy after transplantation. Median CRP level on the day of transplantation was 4.4 (2.8-15.2) mg/L, and 29 patients (33.7%) had a CRP level >5 mg/L.

Characteristics of donors and transplants are given in Table 1. Of note, there were 3 combined transplants (liver, heart, and lung, respectively). The presence of anti-human leukocyte antigen antibodies at the time of transplantation was noted in 44.2% of cases, and the presence of donorspecific antibodies was noted in 11.6%. Nine patients (10.5%) had received immunosuppression previously because of a previous transplant. The immunosuppressive regimen consisted of induction by anti-thymocyte globulin in 41 cases (47.7%) and basiliximab in the remaining 45 cases (53.3%); maintenance therapy included at some point during follow-up corticosteroids (all cases), calcineurin inhibitors (all cases; tacrolimus for 74 patients [91.9%] and cyclosporine for 19 [22.1%]), mycophenolate mofetil (83 cases; 96.5%), azathioprine (11 cases; 12.8%), mammalian target of rapamycin inhibitors (9 cases; 10.5%), and belatacept (5 cases; 5.8%). In addition, 8 patients (9.3%) received rituximab as part of induction therapy or as treatment of an acute rejection episode.

Median duration of follow-up was 55.8 (28.3-86.8) months, for a total of 413.1 person-years.

Overall and Graft Survival

Figure 2A illustrates patient survival. A total of 17 patients died during follow-up, for survival rates of 94.0% (95% confidence interval [CI], 89.1%-99.2%) at 1 year and 85.5% (95% CI, 77.8%-94.0%) at 5 years after transplantation. Causes of death were infection (9 cases; 52.9%), cardiovascular disease (3 cases; 17.6%), cancer (1 case; 5.9%), and unknown (4 cases; 23.5%).

Figure 2B shows the cumulative incidence of graft loss and death (analyzed as competing risks) during follow-up. At 1 year, the cumulative incidence of graft loss was 10.5%



Figure 2. Kaplan-Meier estimate of patient survival (A) and cumulative incidence of mortality and graft loss (B) in patients with AA amyloidosis. Tables represent patients at risk.

(95% CI, 4.0%-17.0%), and at 5 years, it was 13.0% (95% CI, 5.8%-20.1%). All-cause graft failure rates were 16.3% (95% CI, 4.9%-27.8%) at 1 year and 24.2% (95% CI, 9.9%-38.3%) at 5 years after transplantation.

Among functioning grafts, the median estimated glomerular filtration rates were 55 (41-68) mL/min/ 1.73 m^2 at 1 year and 45 (33-65) mL/min/ 1.73 m^2 at 5 years after transplantation.

Amyloidosis Recurrence

During follow-up, a total of 5 histologically confirmed amyloid disease recurrences occurred in allografts (5.8% of the cohort) after a median time of 23.5 (23.1-33.3) months. The main features of these recurrences are presented in Table 2. At recurrence, the median glomerular filtration rate was 28 (26-35) mL/min/1.73 m² and the median UPCR was 0.6 (0.2-1.6) g/g. The diagnosis was made on a protocol 3-month biopsy for 1 patient and on an indicated biopsy for the other 4 patients. Among these patients, mean CRP level during follow-up was 7.8 (5.2-16.3) mg/L.

In comparison, in the whole AA amyloidosis cohort, the median UPCRs were 0.2 (0.1-0.3) g/g at 1 year and 0.1 (0.0-0.2) g/g at 5 years; 2 patients had proteinuria >1 g/g at 1 year, as did 1 patient at 5 years. During follow-up, in the 62 cases with available data, mean CRP level was 6.8

AJKD

				UPCR,	6/g			Time I Intil	aGER at		
Pt. No./Age at KT, y	Cause of Amyloidosis	Specific Therapy ^a	Mean CRP, mg/L ^b	3 mo	+	5 y	At Recurrence	Recurrence, mo	Recurrence, mL/min/1.73 m ²	Location of Deposits	Outcome
1/65	Schnitzler syndrome	Anakinra	2	0.2	0.1	0.1	0.2	n	41	Capsular	Died with functioning graft after 116 mo°
2/46	Unidentified genetic syndrome	None	7	ΥN	0.1	0.0	1.6	78.1	24	Glomerular and vascular	Lost graft after 86 mo; alive at last follow-up (89 mo)
3/60	FMF	Colchicine	171	0.8	0.3	AN	2.3	23.5	26	Glomerular and vascular	Lost graft after 28 mo; died after 33 mo (multivisceral amyloidosis)
4/38	Ankylosing spondylitis	Etanercept	Q	0.1	0.1	0.1	0.2	33.3	35	Interstitial	Alive with functioning graft at last follow-up (101 mo)
5/60	Unknown	None	21	0.6	0.2	AN	0.6	23.1	28	Vascular	Alive with functioning graft at last follow-up (32 mo)
Abbreviations: CF ^a Any anti-inflamm ^b Mean CRP meas ^c Ulterior graft biop	RP, C-reactive protein; atory treatment relative surement including all osy did not show amy	; eGFR, estimated e to the cause of a annual values until loid deposits.	glomerular filtration myloidosis that was recurrence.	rate; FMF, s received a	familial M ífter KT.	lediterrar	iean fever; KT, kidne	ey transplantation; NA	, not available; UPCR, urin	e protein-creatinine ratic	

(3.0-16.0) mg/L; mean CRP level during follow-up was >5 mg/L in 38 cases.

Of note, a total of 102 kidney graft biopsies were performed in 67 patients during follow-up (37 indicated and 65 per protocol, including 43 performed at 3 months and 22 at 1 year after transplantation). For 57 of these biopsies (55.9%), analysis included Congo red stain. In total, 44 patients had at least one kidney graft biopsy that included Congo red stain analysis during follow-up (51.2% of the whole AA amyloidosis cohort) after a median time of 12.0 (3.0-23.4) months.

Acute Rejection Episodes and Infectious, Cardiovascular, and Neoplastic Disease Events

Table 3 summarizes the occurrences of immunologic, infectious, cancer, and cardiovascular events during followup. Twenty-four patients (27.9%) had at least one episode of acute graft rejection (12 cellular, 7 antibodymediated, and 5 mixed) after a median of 4.0 (IQR, 0.9-8.1) months. An infectious episode requiring hospitalization was recorded for 48 patients (55.8%) after a median of 4.8 (IQR, 0.7-20.0) months; the most frequent infections were pyelonephritis (23 cases), pneumonia (14 cases), and intestinal infection (5 cases). Thirteen patients (15.1%) presented with an opportunistic infection: fungal infection, pneumocystis pneumonia, cytomegalovirus disease, varicella, and tuberculosis (n = 7, 2, 2, 1, and 1,respectively). Twenty-two patients (25.8%) experienced an acute cardiovascular event, and 6 (7.0%) were diagnosed with cancer.

The use of biotherapy after transplantation was not associated with the occurrence of an acute rejection episode (4 acute rejection episodes in the 16 patients treated with biotherapy vs 20 in the other 70 patients; P = 0.9, log-rank test) or the occurrence of infection requiring hospitalization (9 occurrences in the 16 patients treated with biotherapy vs 39 in the other 70 patients; P = 0.8, log-rank test).

Factors Associated With Patient Survival

As shown in Fig 3A, there was no significant difference in patient survival according to cause of AA amyloidosis (FMF, other genetic autoinflammatory syndrome, chronic autoinflammatory disease, chronic infection, unknown cause; P = 0.08, log-rank test). Similarly, patient survival was comparable between patients with AA amyloidosis secondary to autoinflammatory diseases (FMF, other autoinflammatory syndromes, and inflammatory diseases) and patients with AA amyloidosis secondary to other causes (infection and unknown origin; P = 0.6, log-rank test; Fig 3B).

Univariable and multivariable analyses using a Cox proportional hazards model were performed to test the associations of different variables with patient survival (Table 4). In the univariable analysis, age at transplantation (hazard ratio [HR], 1.06; 95% CI, 1.02-1.11; P = 0.002), preexisting diabetes (HR, 5.09; 95% CI, 1.44-18.00;

Table 2. Description of AA Amyloidosis Allograft Recurrence Cases

 Table 3.
 Immunologic, Infectious, and Neoplastic Disease and

 Cardiovascular Events During Follow-up in the AA Amyloidosis
 Group

Event	n (%)ª	Time, mo (IQR) ^b
Cellular acute graft rejection	17 (19.8%)	4.7 (1.0-6.3)
Antibody-mediated acute graft rejection	12 (14.0%)	6.6 (2.1-24.5)
Acute graft rejection (cellular or antibody-mediated)	24 (27.9%)	4.0 (0.9-8.1)
Infection requiring hospitalization	48 (55.8%)	4.8 (0.7-20.0)
Opportunistic infection	13 (15.1%)	22.3 (5.2-27.5)
Cardiovascular event	22 (25.8%)	36.2 (15.6-59.8)
Cancer	6 (7.0%)	36.1 (26.4-53.9)

^aNumber of transplants that presented at least one episode.

^bMedian time to onset of the first episode.

P = 0.01), and CRP level at transplantation (HR, 1.01; 95% CI, 1.00-1.02; P = 0.02) were associated with patient survival. In the multivariable model, age at transplantation (HR, 1.10; 95% CI, 1.03-1.18; P = 0.007), preexisting diabetes (HR, 10.62; 95% CI, 1.94-58.05; P = 0.006), and CRP level at transplantation (HR, 1.01; 95% CI, 1.00-1.02; P = 0.01) remained statistically associated with patient survival.

Factors Associated With Graft Survival

Cause of AA amyloidosis (FMF, other autoinflammatory syndrome, chronic autoinflammatory disease, chronic infection, unknown cause) was not associated with graft survival (P = 0.9, Gray's test). Graft survival was similar between patients with AA amyloidosis secondary to auto-inflammatory diseases (FMF, other genetic syndromes, and inflammatory diseases) and those with AA amyloidosis secondary to other causes (infection and unknown origin; P = 0.6, Gray's test).

Univariate and multivariate analysis using a causespecific hazard model were performed to test the associations of different variables with graft survival (Table 5). In the univariate analysis, only the presence of donor-specific antibodies at the time of transplantation was statistically associated with graft survival (HR, 4.57; 95% CI, 1.13-18.40; P = 0.03). In the multivariate model, donor-specific antibodies at the time of transplantation (HR, 9.59; 95% CI, 2.28-40.35; P = 0.002), CRP level at the time of transplantation (HR, 1.68; 95% CI, 1.10-2.57; P = 0.02), and cold ischemia time (HR, 0.99; 95% CI, 0.99-1.00; P = 0.04) were statistically associated with graft survival.

Discussion

To our knowledge, the present study is the largest cohort to report outcomes of kidney transplantation in patients with renal AA amyloidosis. Our main findings are high overall and graft survival rates and a low amyloidosis recurrence rate, suggesting that patients with AA amyloidosis can benefit from kidney transplantation with favorable outcomes.

As a mean of comparison in the absence of a control group for our cohort, data for 34,776 patients from the French transplantation database (Agence de la Biomédecine, rapport Rein 2018²⁵) between 2007 and 2017 demonstrate a patient survival rate of 88.4% at 5 years (95% CI, 88.0-89.0) and an all-cause graft survival rate of 78.6% (95% CI, 78.1-79.1). During almost the same time period, in our cohort, the patient survival rate was slightly lower but comparable (85.5% [95% CI, 77.8%-94.0%] at 5 years). The overall graft survival rate in our cohort was 75.8% (95% CI, 61.8%-90.1%) at 5 years, which is also similar to national data. Interestingly, Kofman et al previously reported a 5-year patient survival rate of 82.5%⁸ for patients who received transplants between 1981 and 2009 that is similar to the findings in our study, which is somewhat surprising given the improvements in posttransplantation global care. This probably reflects that the overall survival of kidney transplant recipients did not improve during this time period in France (the 5-year survival rate of patients who received transplants between 1996 and 2000 was 92.4%, vs 87.8% between 2012 and 2014), an outcome we attribute to kidney recipients being older and having more comorbidities in recent years than in earlier eras. Indeed, in the study by Kofman et al, the mean age at transplantation was 44.1 years, compared with a median of 49.4 years in the present study.

In contrast, most previously published series of kidney transplant recipients with AA amyloidosis reported poorer patient survival compared with controls: the historic series of Pasternack et al reported a patient survival rate of approximately 50% at 3 years (vs 79% in controls with glomerulonephritis, n = 45).⁹ Similarly, Kofman et al reported a 5-year survival rate of 82.5%, versus 94.2% for the matched control group (P = 0.028).⁸ A Turkish study focusing exclusively on AA amyloidosis secondary to FMF that included 81 kidney transplant recipients also found lower survival rated compared with controls without amyloidosis (92.5% vs 100% at 5 years; P = 0.02).¹⁴ Finally, Law et al recently reported a survival rate of AA amyloidosis kidney transplant recipients (N = 48) of 84% at 5 years, which was comparable to that in controls with diabetic nephropathy but lower than that in controls with autosomal-dominant polycystic kidney disease.¹⁵ The 5year patient survival rate of 85.5% in our cohort is comparable to, or even a bit better than, outcomes in most of the series cited here. However, it should be noted that the data from these earlier studies are heterogenous, and, in particular, transplant periods span a very long time, with all studies including transplants from before 1990. Our hypothesis that better pre- and posttransplant control of the disease that causes AA amyloidosis in the more recent era would lead to a higher survival could not be verified, even though there might be a trend toward better results. In terms of potential consequences of the hypotension that





Figure 3. Kaplan-Meier estimates of patient survival in patients with AA amyloidosis according to the cause of amyloidosis. (A) Patients are separated into 5 main causes of AA amyloidosis. (B) The autoinflammatory disease group includes genetic autoinflammatory syndromes and chronic inflammatory diseases. Tables represent patients at risk in each group. Statistical difference was assessed with the log-rank test. Abbreviation: FMF, familial Mediterranean fever.

	Alive	Dead	Univariate Analysis	;	Multivariate Analysis	Multivariate Analysis		
Factor	(n = 69)	(n = 17)	HR (95% CI)	Р	HR (95% CI)	Р		
Age at KT, y	47.2 (37.9-58.8)	61.3 (56.2-65.6)	1.06 (1.02-1.11)	0.002	1.10 (1.03-1.18)	0.007		
Duration of dialysis before KT, mo	39 (20-62)	40 (18-77)	1.01 (0.99-1.02)	0.2	1.01 (0.99-1.02)	0.3		
Preexisting diabetes	0	3 (17.6%)	5.09 (1.44-18.00)	0.01	10.62 (1.94-58.05)	0.006		
CRP at KT, mg/L	4.5 (2.7-13.3)	4.3 (3.1-21.5)	1.01 (1.00-1.02)	0.02	1.01 (1.00-1.02)	0.01		
Each factor was analyzed in	dividually (universitate analy	ucia) hafara narfarming a m	ultivariata analysia (Cay maa	المعارك والمالي	ianas CDD C reactive proteins	KT kidnow		

 Table 4.
 Association of Different Factors With Patient Survival in AA Amyloidosis KT Recipients

Each factor was analyzed individually (univariate analysis) before performing a multivariate analysis (Cox model). Abbreviations: CRP, C-reactive protein; KT, kidney transplantation.

is classically associated with amyloidosis, 5.8% of the present cohort experienced primary nonfunction and 27.9% had delayed graft function. Even if these rates are higher than usually observed, the association with AA amyloidosis is uncertain because we do not have a record of measured blood pressure during follow-up.

Another important finding of our study is the low rate of AA amyloidosis recurrence in the graft compared with other studies: we found an incidence of 5.8%, whereas most other studies reported rates ranging from 8%^{9,26} to 26%.¹⁴ In the largest series, the rates were 14%,⁸ 19%,^{13,15} and 26%.¹⁴ Again, we hypothesize that our low recurrence rate is a result of a more recent transplantation period, in the era of biotherapies, enabling better control of the underlying inflammatory disease. However, these different recurrence rates may be explained by differing attitudes toward histological monitoring during follow-up. Unfortunately, none of the cited studies detail the number of biopsies performed during follow-up. In the present study, a high number of biopsies was performed (65 protocol biopsies), but the diagnosis of subclinical amyloidosis recurrence may have been underestimated, as only 44 patients (51.2%) underwent at least one kidney biopsy with Congo red stain analysis. However, very few patients had proteinuria during follow-up (only 2 patients had a UPCR >1 g/g at 1 year after transplantation, along with 1 at 5 years), which suggests that clinically significant recurrences were not underdiagnosed. Because of this low recurrence rate, the present study lacked the statistical power to explore an association between amyloidosis recurrence and patient or graft survival and to identify factors associated with disease recurrence.

In contrast, we observed a high occurrence of acute graft rejection episodes in the cohort: 27.9% of patients with AA amyloidosis experienced at least one cellular or antibody-mediated acute rejection episode during followup. Our hypothesis is that the concern for infection in these potentially susceptible patients led to lower dosing of immunosuppressive agents.

Another important finding of the present study was the association of CRP concentration with survival outcomes. Indeed, a high CRP concentration at the time of transplantation was linked to poorer patient and graft survival rates on multivariate analysis. To our knowledge, ours is the first study to report such an association. Indeed, Pinney et al showed an association between median SAA concentration during follow-up and the recurrence of amyloid on the graft.¹³ Similarly, Sarihan et al showed higher CRP levels during follow-up in patients with amyloid recurrence.¹⁴ However, none of the large series reported a significant association of SAA or CRP levels with patient or graft survival.^{8,13-15} This association with patient and graft survival again highlights the importance of controlling the underlying inflammatory disease before transplantation. In general, we believe careful selection before transplantation is particularly important for patients with AA amyloidosis; it should include a screening of other organs' involvement and consider the underlying disease activity.

Biotherapies are a focus of interest for this purpose. In the present study, only 18.6% of the amyloidosis cohort received biotherapy after transplantation (anti–IL-1, anti–IL-6, or anti–TNF- α). The recent series by Law et al is the only other study of AA amyloidosis kidney transplant recipients to include patients receiving biotherapies: 33.3% of their cohort (n = 16) received such treatments after transplantation.¹⁵ Other small series have illustrated that anti–IL-1 therapies can be used safely in the transplantation setting.²⁷⁻³¹ A recent review of cases reported 37 kidney transplant recipients undergoing anti–IL-1

Table 5.	Association	of Different	Factors W	lith Graft	Survival in	AA Am	yloidosis	KT Rec	ipients
----------	-------------	--------------	-----------	------------	-------------	-------	-----------	--------	---------

	Functioning	Graft Loss	Univariate Analysis		Multivariate Analysi	s
	Graft (n = 71)	(n = 15)	HR (95% CI)	Р	HR (95% CI)	Р
Age of donor, y	53 (38-65)	52 (44-63)	1 (0.97-1.04)	0.9	1.02 (0.98-1.05)	0.4
Cold ischemia, min	911 (735-1,263)	810 (579-995)	0.99 (0.99-1.00)	0.1	0.99 (0.99-1.00)	0.04
DSA at KT	7 (9.9%)	3 (20.0%)	4.57 (1.13-18.40)	0.03	9.59 (2.28-40.35)	0.002
CRP at KT, mg/L	4.2 (2.6-12.2)	12.8 (3.0-45.5)	1.47 (0.98-2.18)	0.06	1.68 (1.10-2.57)	0.02

Each factor was analyzed individually before performing a multivariate analysis (cause-specific hazards model). Data are presented as n (%) or median (IQR) in each group. The HR is presented with its 95% CI and corresponding *P* value. Abbreviations: CI, confidence interval; CRP, C-reactive protein; DSA, donor-specific antibodies; HR, hazard ratio; KT, kidney transplantation. therapy and 22 undergoing anti–TNF- α therapy, with low infection and rejection rates³² suggesting that biotherapies are safe to use in this setting. The retrospective nature of our study and relatively infrequent use of biotherapies limits the ability to analyze the effect of these treatments on kidney transplant outcomes. More studies are needed to investigate these aspects.

In conclusion, the present study found that AA amyloidosis kidney transplant recipients experienced favorable survival outcomes and lower recurrence rates than previously reported. These data support the practice of treating these patients with kidney transplantation for end-stage kidney disease.

Article Information

Authors' Full Names and Academic Degrees: Chloë Schwarz, MD, Sophie Georgin-Lavialle, MD, PhD, Yannis Lombardi, MD, Olivier Marion, MD, Frédéric Jambon, MD, Christophe Legendre, MD, PhD, David Marx, MD, Charlène Levi, MD, Fatouma Toure, MD, PhD, Moglie Le Quintrec, MD, PhD, Mickael Bobot, MD, Marie Matignon, MD, Amaury Dujardin, MD, Mehdi Maanaoui, MD, Sébastien Cuozzo, MD, Arwa Jalal-Eddine, MD, Kévin Louis, MD, PhD, Inna Mohamadou, MD, François Brazier, MD, Tristan De Nattes, MD, Claire Geneste, MD, Eric Thervet, MD, PhD, Didier Ducloux, MD, PhD, Valentin Mayet, MD, Raphaël Kormann, MD, PhD, Antoine Lanot, MD, Agnès Duveau, MD, PhD, Mohamad Zaidan, MD, PhD, Laurent Mesnard, MD, PhD, Nacera Ouali, MD, Eric Rondeau, MD, PhD, Camille Petit-Hoang, MD, Vincent Audard, MD, PhD, Aurélie Deshayes, MS, Anissa Moktefi, MD, PhD, Marion Rabant, MD, PhD, David Buob, MD, PhD, Hélène François, MD, PhD, and Yosu Lugue, MD, PhD.

Authors' Affiliations: Sorbonne Université, Institut National de la Santé et de la Recherche Médicale, Unité Mixte de Recherche 1155, Soins Intensifs Néphrologiques et Rein Aigu, Département de Néphrologie, Assistance Publique-Hôpitaux de Paris, Hôpital Tenon (CS, YLombardi, LM, NO, ER, CP-H, HF, YLuque), Sorbonne Université, Internal Medicine Department, Assistance Publique-Hôpitaux de Paris, Hôpital Tenon, National Reference Center for Autoinflammatory Diseases and Inflammatory Amyloidosis, Groupe de recherche clinique Amylose AA Sorbonne Université (GRAASU) (SG-L), Université de Paris, Service de Néphrologie-Transplantation (CLegendre) and Pathology Department (MR), Hôpital Necker, Assistance Publique-Hôpitaux de Paris, Nephrology and Transplantation Department, Saint-Louis Hospital, Assistance Publique-Hôpitaux de Paris (KL), Kidney Transplantation Department, Hôpital Pitié-Salpétriêre, Sorbonne Université, Assistance Publique-Hôpitaux de Paris (IM). Nephrology Department, European Georges Pompidou Hospital, Assistance Publique-Hôpitaux de Paris (ET), and Pathology Department, Sorbonne Université, Tenon Hospital, Assistance Publique-Hôpitaux de Paris (DB), Paris; Université de Paris, Service de Néphrologie-Transplantation, Assistance Publique-Hôpitaux de Paris, Hôpital Bicêtre, Le Kremlin-Bicêtre (CS, MZ); Department of Nephrology and Organ Transplantation, Toulouse Rangueil University Hospital, Toulouse (OM); Centre Hospitalier Universitaire de Bordeaux, Service de Néphrologie, Transplantation Dialyse, Aphérèses, Hôpital Pellegrin, Place Amélie Raba Léon, Bordeaux (FJ); Department of Nephrology and Transplantation, Strasbourg University Hospital, Strasbourg (DM); Service de Transplantation Rénale, Hôpitaux Civils, Lyon (CLevi); Department of Nephrology, Dialysis and Transplantation, Hospital University of Limoges, Limoges (FT); Service de Transplantation Rénale, Centre Hospitalier Universitaire Montpellier, Montpellier (MLQ); Centre de Néphrologie et Transplantation Rénale,

Assistance Publique-Hôpitaux de Marseille, Hôpital de la Conception, Centre Hospitalier Universitaire de la Conception, Marseille (MB); Nephrology and Renal Transplantation Department, Assistance Publique-Hôpitaux de Paris, Hôpital Henri Mondor, Creteil (MMatignon, VA); Service de Néphrologie et Immunologie Clinique, Nantes Université, Centre Hospitalier Universitaire Nantes, Institut National de la Santé et de la Recherche Médicale, Centre de Recherche en Transplantation et Immunologie, Unité Mixte de Recherche 1064, Institut de Transplantation Urologie Néphrologie, Nantes (ADujardin); Nephrology Department, Centre Hospitalier Universitaire de Lille, Lille (MMaanaoui); Department of Nephrology Dialysis and Transplantation, Pasteur 2 Hospital, Nice University Hospital, Nice (SC); Department of Nephrology, Foch Hospital, Suresnes (AJ-E); Nephrology, Internal Medicine, Dialysis and Transplantation Department, Centre Hospitalier Universitaire Amiens, Amiens (FB); Nephrology-Hemodialysis Department, Centre Hospitalier Universitaire Rouen, Rouen (TDN); Nephrology Department, Centre Hospitalier Universitaire Tours, Tours (CG); Nephrology Department, Centre Hospitalier Universitaire Besançon, Besançon (DD); Nephrology-Hemodialysis Department, Centre Hospitalier Universitaire Clermont-Ferrand, Clermont-Ferrand (VM); Nephrology Department, Centre Hospitalier Universitaire Nancy, Nancy (RK); Nephrology-Dialysis-Kidney Transplantation Department, Centre Hospitalier Universitaire Caen, Caen (AL); Nephrology Department, Centre Hospitalier Universitaire Angers, Angers (ADuveau); Agence de la Biomédecine, St Denis (ADeshayes); and Pathology Department, Henri Mondor Hospital, Assistance Publique-Hôpitaux de Paris, Créteil (AM), France.

Address for Correspondence: Yosu Luque, MD, PhD, Soins Intensifs et Rein Aigu, Département de Néphrologie, Hôpital Tenon, 4 rue de la Chine, 75020 Paris, France. Email: yosu. Iuque@aphp.fr

Authors' Contributions: Research idea and study design: YLuque, HF, SG-L, DB, VA, CS; data acquisition: CS, OM, FJ, CLegendre, DM, CLevi, FT, MLQ, MB, MMatignon, ADujardin, MMaanaoui, SC, AJ-E, KL, IM, FB, TDN, CG, ET, DD, VM, RK, AL, ADuveau, MZ, AM, MR, ADeshayes; data analysis/interpretation: CS, YLuque, SG-L, HF, DB; statistical analysis: CS, YLuque, YLombardi; supervision and mentorship: YLuque, HF, SG-L, DB, VA, LM, NO, CPH, ER. Each author contributed important intellectual content during manuscript drafting or revision and agrees to be personally accountable for the individual's own contributions and to ensure that questions pertaining to the accuracy or integrity of any portion of the work, even one in which the author was not directly involved, are appropriately investigated and resolved, including with documentation in the literature if appropriate.

Support: None.

Financial Disclosure: The authors declare that they have no relevant financial interests.

Acknowledgments: We thank Emilie Savoye (Agence de la Biomédecine) for the statistical expertise.

Data Sharing: A subset of deidentified data may be available upon request to the corresponding author.

Peer Review: Received September 30, 2022. Evaluated by 2 external peer reviewers, with direct editorial input from a Statistics/ Methods Editor, an Associate Editor, and the Editor-in-Chief. Accepted in revised form July 24, 2023.

References

 Benson MD, Buxbaum JN, Eisenberg DS, et al. Amyloid nomenclature 2018: recommendations by the International Society of Amyloidosis (ISA) nomenclature committee. *Amyloid.* 2018;25(4):215-219. doi:10.1080/13506129.2018. 1549825

- Pinney JH, Smith CJ, Taube JB, et al. Systemic amyloidosis in England: an epidemiological study. Br J Haematol. 2013;161(4):525-532. doi:10.1111/bjh.12286
- 3. Magy-Bertrand N, Dupond JL, Mauny F, et al. Incidence of amyloidosis over 3 years: the AMYPRO study. *Clin Exp Rheumatol.* 2008;26(6):1074-1078.
- Winburn I, Ishii T, Sumikawa T, Togo K, Yasunaga H. Estimating the prevalence of transthyretin amyloid cardiomyopathy in a large in-hospital database in Japan. *Cardiol Ther.* 2019;8(2): 297-316. doi:10.1007/s40119-019-0142-5
- Lachmann HJ, Goodman HJB, Gilbertson JA, et al. Natural history and outcome in systemic AA amyloidosis. N Engl J Med. 2007;356(23):2361-2371. doi:10.1056/NEJMoa070265
- Bergesio F, Ciciani AM, Manganaro M, et al. Renal involvement in systemic amyloidosis: an Italian collaborative study on survival and renal outcome. *Nephrol Dial Transplant*. 2008;23(3): 941-951. doi:10.1093/ndt/gfm684
- Bollée G, Guery B, Joly D, et al. Presentation and outcome of patients with systemic amyloidosis undergoing dialysis. *Clin J Am Soc Nephrol.* 2008;3(2):375-381. doi:10.2215/CJN.02470607
- Kofman T, Grimbert P, Canouï-Poitrine F, et al. Renal transplantation in patients with AA amyloidosis nephropathy: results from a French multicenter study. *Am J Transplant*. 2011;11(11): 2423-2431. doi:10.1111/j.1600-6143.2011.03620.x
- Pasternack A, Ahonen J, Kuhlbäck B. Renal transplantation in 45 patients with amyloidosis. *Transplantation*. 1986;42(6):598-601. doi:10.1097/00007890-198612000-00005
- Celik A, Saglam F, Dolek D, et al. Outcome of kidney transplantation for renal amyloidosis: a single-center experience. *Transplant Proc.* 2006;38(2):435-439. doi:10.1016/j.transproceed.2006.01.003
- 11. Haq A, Hussain S, Meskat B, Mohan P, Conlon P, Hickey DP. Complications of renal transplantation in patients with amyloidosis. *Transplant Proc.* 2007;39(1):120-124. doi:10. 1016/j.transproceed.2006.10.021
- Gursu M, Yelken B, Caliskan Y, et al. Outcome of patients with amyloidosis after renal transplantation: a single-center experience. *Int J Artif Organs*. 2012;35(6):444-449. doi:10.5301/ ijao.5000091
- Pinney JH, Lachmann HJ, Sattianayagam PT, et al. Renal transplantation in systemic amyloidosis-importance of amyloid fibril type and precursor protein abundance. *Am J Transplant.* 2013;13(2):433-441. doi:10.1111/j.1600-6143.2012.04326.x
- Sarihan I, Caliskan Y, Mirioglu S, et al. Amyloid A amyloidosis after renal transplantation: an important cause of mortality. *Transplantation*. 2020;104(8):1703-1711. doi:10.1097/TP. 000000000003043
- Law S, Cohen O, Lachmann HJ, et al. Renal transplant outcomes in amyloidosis. *Nephrol Dial Transplant*. 2021;36(2): 355-365. doi:10.1093/ndt/gfaa293
- Zemer D, Pras M, Sohar E, Modan M, Cabili S, Gafni J. Colchicine in the prevention and treatment of the amyloidosis of familial Mediterranean fever. *N Engl J Med.* 1986;314(16): 1001-1005. doi:10.1056/NEJM198604173141601
- Livneh A, Zemer D, Siegal B, Laor A, Sohar E, Pras M. Colchicine prevents kidney transplant amyloidosis in familial Mediterranean fever. *Nephron.* 1992;60(4):418-422. doi:10. 1159/000186801

- Ozen S, Demirkaya E, Erer B, et al. EULAR recommendations for the management of familial Mediterranean fever. *Ann Rheum Dis.* 2016;75(4):644-651. doi:10.1136/annrheumdis-2015-208690
- Fernández-Nebro A, Olivé A, Castro MC, et al. Long-term TNFalpha blockade in patients with amyloid A amyloidosis complicating rheumatic diseases. *Am J Med.* 2010;123(5):454-461. doi:10.1016/j.amjmed.2009.11.010
- Kuroda T, Wada Y, Kobayashi D, et al. Effective anti-TNFalpha therapy can induce rapid resolution and sustained decrease of gastroduodenal mucosal amyloid deposits in reactive amyloidosis associated with rheumatoid arthritis. *J Rheumatol.* 2009;36(11):2409-2415. doi:10.3899/ jrheum.090101
- Fishman JA. Infection in solid-organ transplant recipients. *N Engl J Med.* 2007;357(25):2601-2614. doi:10.1056/ NEJMra064928
- Bertrand D, Chavarot N, Gatault P, et al. Opportunistic infections after conversion to belatacept in kidney transplantation. *Nephrol Dial Transplant*. 2020;35(2):336-345. doi: 10.1093/ndt/gfz255
- Stankovic Stojanovic K, Hentgen V, Fellahi S, et al. Concordance between CRP and SAA in familial Mediterranean fever during attack-free period: a study of 218 patients. *Clin Biochem.* 2017;50(4-5):206-209. doi:10.1016/j.clinbiochem.2016.11.008
- Legger GE, Dermer CWE, Brunger AF, van Daele PLA, Nienhuis HLA. The relation between C-reactive protein and serum amyloid A in patients with autoinflammatory diseases. *Pediatr Rheumatol Online J.* 2022;20(1):106. doi:10.1186/ s12969-022-00757-9
- Agence de la Biomédecine. Rapport médical et scientifique 2018. Accessed September 12, 2020. https://rams.agencebiomedecine.fr/
- Altindal M, Turkmen E, Yildirim T, et al. Kidney transplantation for end-stage renal disease secondary to familial Mediterranean fever. *Clin Transplant*. 2016;30(7):787-790. doi:10.1111/ctr.12749
- Moser C, Pohl G, Haslinger I, et al. Successful treatment of familial Mediterranean fever with Anakinra and outcome after renal transplantation. *Nephrol Dial Transplant*. 2009;24(2): 676-678. doi:10.1093/ndt/gfn646
- Özçakar ZB, Keven K, Çakar N, Yalçınkaya F. Transplantation within the era of anti-IL-1 therapy: case series of five patients with familial Mediterranean fever-related amyloidosis. *Transpl Int.* 2018;31(10):1181-1184. doi:10.1111/tri.13312
- Yildirim T, Yilmaz R, Uzerk Kibar M, Erdem Y. Canakinumab treatment in renal transplant recipients with familial Mediterranean fever. *J Nephrol.* 2018;31(3):453-455. doi:10.1007/ s40620-018-0475-5
- Sendogan DO, Saritas H, Kumru G, et al. Outcomes of canakinumab treatment in recipients of kidney transplant with familial Mediterranean fever: a case series. *Transplant Proc.* 2019;51(7): 2292-2294. doi:10.1016/j.transproceed.2019.03.049
- Trabulus S, Korkmaz M, Kaya E, Seyahi N. Canakinumab treatment in kidney transplant recipients with AA amyloidosis due to familial Mediterranean fever. *Clin Transplant*. 2018;32(8):e13345. doi:10.1111/ctr.13345
- 32. Yazısız V, Yılmaz VT, Uçar İ, et al. The use of anti-interleukin-1 agents and tumor necrosis factor-alpha inhibitors in renal transplant recipients. *Arch Rheumatol.* 2021;36(3):366-374. doi:10.46497/ArchRheumatol.2021.8500